Author's response to reviews

Title: Promoting E2F1-mediated apoptosis in oestrogen receptor-alpha-negative breast cancer cells

Authors:

Maria F Montenegro (fermontenegro@um.es)
Maria del Mar Collado-Gonzales (mdmcq1@um.es)
Maria Piedad Fernandez-Perez (mfp18858@um.es)
Manel B Hammouda (ma_nel85@hotmail.com)
Lana Tolordava (lanatolordava@mail.ru)
Mariam Gamkrelidze (gamkrelidzem@yahoo.com)
Jose Neptuno Rodriguez-Lopez (neptuno@um.es)

Version: 4
Date: 7 July 2014

Author's response to reviews: see over
Author’s response to reviews

Title: Promoting E2F1-mediated apoptosis in oestrogen receptor-α-negative breast cancer cells

Authors:

María F. Montenegro (fermontenegro@um.es)
María del Mar Collado-González (mdmcg1@um.es)
María Piedad Fernández-Pérez (mfp18858@um.es)
Manel B. Hammouda (ma_nel85@hotmail.com)
Lana Tolordava (lanatolordava@mail.ru)
Mariam Gamkrelidze (gamkrelidzem@yahoo.com)
José Neptuno Rodríguez-López (neptuno@um.es)

Version: 4 Date: 07 July 2014

Author’s response to reviews: see over
Dear Editor,

Thank you for your letter concerning to our manuscript entitles “Promoting E2F1-mediated apoptosis in oestrogen receptor-α-negative breast cancer cells” (ID: 2083042115115977). The original manuscript was edited with the help of a language editing service; however, we agree with you that some grammatical errors were introduced during the last revision of the manuscript. Now, we have corrected the manuscript with the help of a native English speaker. We hope that the changes made in the new version will enable you to expedite publication.

Sincerely yours

Prof. Jose Neptuno Rodriguez Lopez
Department of Biochemistry and Molecular Biology A
University of Murcia
30100 Murcia (Spain)
Tel. +34 868 888284
Fax. +34 868 884147
e-mail address: neptuno@um.es